Author's response to reviews

Title: Association of the rs738409 polymorphism in PNPLA3 with liver damage and the development of nonalcoholic fatty liver disease

Authors:

Kikuko Hotta (kikukoh@kuhp.kyoto-u.ac.jp)
Masato Yoneda (yoneda@med.yokohama-cu.ac.jp)
Hideyuki Hyogo (hidehyogo@hiroshima-u.ac.jp)
Hidenori Ochi (hochi@hiroshima-u.ac.jp)
Seiho Mizusawa (mizusawa@genome.med.kyoto-u.ac.jp)
Takato Ueno (takato@med.kurume-u.ac.jp)
Kazuaki Chayama (chayama@mba.ocn.ne.jp)
Atsushi Nakajima (nakajima-ty@umin.ac.jp)
Kazuwa Nakao (nakao@kuhp.kyoto-u.ac.jp)
Akhiro Sekine (sekine@genome.med.kyoto-u.ac.jp)

Version: 4 Date: 20 December 2010

Author's response to reviews: see over
20 December 2010

The BMC Series Editorial Production Team

Dear Sir/Madam;

On behalf of all the authors, I am sending the proofreading MS:1989300951439226, entitled "Association of the rs738409 polymorphism in PNPLA3 with liver damage and the development of nonalcoholic fatty liver disease".

1. We have changed “Supplementary Table 1 or 2” to “Table S1 or S2” in the text.

2. We have changed all text in black throughout the text.

3. We have corrected the format of Tables.

4. We have corrected some typographical errors.

Sincerely,

Kikuko Hotta, M.D., Ph.D.
Assistant Professor
EBM Research Center,
Kyoto University Graduate School of Medicine
Yoshida-Konoecho, Sakyo-ku, Kyoto
606-8501 Japan
Phone: +81-75-751-2022
Fax: +81-75-751-2082
kikukoh@kuhp.kyoto-u.ac.jp